Author's response to reviews

Title: Hypoglycemia mediated by paraneoplastic production of Insulin like Growth Factor - 2 by malignant renal solitary fibrous tumor - clinical case and literature review

Authors:

Ameer Khowaja (khowa002@umn.edu)
Brianna Johnson-Rabbett (joh08932@umn.edu)
John Bantle (bantl001@umn.edu)
Amir Moheet (mohee002@umn.edu)

Version: 2
Date: 9 December 2013

Author's response to reviews:

Revision 1: Title Page
Names of all authors, institutional affiliations, institutional address with countries and email addresses included. Furthermore postal address of submitting/corresponding author also included.

Revision 2: Acknowledgements
No acknowledgement is required. Section of acknowledgement added on page 12 with comment "None".